Author’s response to reviews

Title: A successfully thrombolysed acute inferior myocardial infarction due to type A aortic dissection with lethal consequences: the importance of early echocardiography.

Authors:

Grigorios Tsigkas (gregtsig@hotmail.com)
Georgios Kasimis (gksup@yahoo.gr)
Konstantinos Theodoropoulos (ktheod2005@hotmail.com)
Konstantinos Chouchoulis (kchouchoulis@hotmail.com)
Nikolaos G Baikoussis (ngbaik@yahoo.com)
Efstratios Apostolakis (stratisapostolakis@yahoo.gr)
Eleni Bousoula (elena_b11@hotmail.com)
Athanasios Moulias (moulias.ath@windowslive.com)
Dimitrios Alexopoulos (dalex@med.upatras.gr)

Version: 4 Date: 20 August 2011

Author’s response to reviews: see over
COVER LETTER

Dear Editor these are the following changes that we made:

1) We used * for corresponding author rather than the image of an x that has been initially included

2) We put the abstract in the manuscript, after the title page

3) We removed all forms of tracking and checked that all text in the manuscript is black and finally we checked the manuscript for any typographical errors.

Address for correspondence: Georgios Kasimis
Cardiologist
Patras University Hospital
Rion Patras 26500 Greece
Tel: +302610999281
Fax: +302610992941
E-mail: gksup@yahoo.gr